
KAISER HILL 
C O M P A N Y  

MEMORANDUM 

Date: November 3, 2004 

To: Jan Robbins, T130G 

From: Randy Leitner 

Subject: WEEKLY STATUS MEETING DOE/LRA for November 3,2004 

Attached are the minutes for the referenced weekly meeting for the Operating Record. 
In attendance were: 

M. Brown 
D. Coyne 
T. Dieter 
S. Garcia 
C. Gilbreath 
H. Gilpin 
B. Larson 
R. Leitner 

S. McNitt 
D. Onyskiw 
J. Parkin 
D. Rack 
W. Seyfert 
R. Vonfeldt 

Additional items discussed: 

Actions Items: 

End 
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38 1AD 1 17BD00 
40 lAD138BD00 
52 lAD140BD00 
Area AA 

D&D Summary ( ACC/ADD) 
Weekly Meeting Date 11/03/04 

GB307 removed. 
0 

Set scope complete. 
Criticality drain removal in basement continues. 

Removal of demisters and 1" stage filters complete. 
Removing Zone 1 duct Rooms 11 15, 1105,1103, and 11 17. 

Set 
6 lAQ31DR02B 

Removing Zone 1 duct Rooms 2325 and 2327.' . 

Construction Status Dan C o p e  

9 (CSV) 

10 1AD110BD00 

Environmental Status 

D&D-Jim Floerke Environmental 

Area AD 

19 Sludge Removal 

RCRA stable - no insps. 

0 

D 

D 

D 

I 

Washington C 
Sets 1 and 2: Electrical and mechanical (piping/supports) removal. 
Set 6: Dry decon in rooms 3549. 
Set 12 : Mechanical Removal (room 110311 115 dudpiping). Continued dry 
decon in room 1 1 17 
Set 13: No activity. 
Set 14: On hold 
Set 17: Dry decon in room 3523 (embed plates). 
Set 18: Mechanical removal (duct, misc. supports & piping). Cont'd drywall 
removal. Continued dry decon in room 3801. 
Set 19: Tank removal complete. 
Set 2 1: Wall removal in room 380914802 complete. 
Set 22: Abandoned electrical, duct, and piping removal. Completed removal of 
SAUs 301/302/303 and preparing FP-321 for removal 

D&D Waste - 
November 2 ,2004 Inventory: 70 drums require handling (2 overmass drums) 

I RCRA Stable. 

OUD Activities 
Set 23: Placed on hold, crew re-assigned to Bldg. 374. 
Set 38: Dismantling of SAU-203 placed on hold to re-direct crew to FP-242. 
Set 40: No activity. 
Set 41 : Plenum FP-242 removal completed. Commence FP-24 1 plenum 
removal. 
Set 57: Continued room 48 14 platform removal. 
Set 58: Continued wall removal in room 4805. 
Temporary electrical power installation on-going in both bldgs. 
Preparing for ACM removal on 374 roof (gravel & insulation removal). 

Randy Leitner 
Net Drum Loss for Week 65 
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D&D Summary ( ACC/ADD) 
Weekly Meeting Date 11/03/04 

PDSR - Steve McNitt 
B371 Total Scan m2 Required Scan m2 Completed I B374 Total Scan m2 Required Scan m2 Completed 
Class I 
Class I1 
Class 111 

13,141 
13,506 
1,632 

0 
0 

88 

Class I 1,845 0 
Class I1 3,437 363 
Class I11 126 53 

Room 3813 & B374 Exterior PDSR approved by CDPHE on 11-2-04. 
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